Reviewer #1:

Comments to the Author

1. *The manuscript meets the requirements of the World Journal of Clinical Cases and, in my opinion, does not require additions. Ethical requirements are met. The language certificate corresponds to the manuscript.*

**Reply:** Thanks a lot for your careful review work and special thanks for your comments.

Reviewer #2:

Comments to the Author

1. *Although primary adenoid cystic carcinoma in the trachea is the 2nd most common tumor of the trachea, this case report may contribute to the literature since it was diagnosed by transesophageal biopsy and presented well.*

**Reply:** Thank you very much for your kindly comments. We also hope that our manuscript could contribute to the literature.